Author's response to reviews

Title: Alcohol consumption in Estonia and Finland: Finbalt survey 1994-2006

Authors:

Kersti Pärna (kersti.parna@ut.ee)
Kaja Rahu (kaja.rahu@tai.ee)
Satu Helakorpi (satu.helakorpi@thl.fi)
Mare Tekkel (mare.tekkel@tai.ee)

Version: 2 Date: 9 December 2009

Author's response to reviews:

Melissa Norton,
Editor-in Chief, BMC Public Health

Re: Revised manuscript 1488007606273069

Dear Melissa Norton, December 09, 2009

Please find attached revised manuscript Alcohol consumption in Estonia and Finland: Finbalt survey 1994–2006. This version was approved by all the authors.

We are very grateful to reviewers for their helpful comments. We hereby detail the changes made to our manuscript in response to the reviewers' comments.

Looking forward to hear from you,

Yours sincerely,

Kersti Pärna (corresponding author)
Department of Public Health
University of Tartu
Ravila 19
50411 Tartu
Estonia
Tel +372 7 374 193